Author's response to reviews

Title: Acquired craniomeningocele in an infant with craniosynostosis, case report.

Authors:

Mostafa El Khashab MD (mostkash@yahoo.com)
Farideh Nejat MD (Nejat@sina.tums.ac.ir)
Shahrooz Yazdani MD (Yazdani@yahoo.com)
Nima Baradaran MD (baradaran.nima@gmail.com)

Version: 4 Date: 18 April 2009

Author's response to reviews:

Dear Editor

We did all necessary changes. Actually there was no errors in the text. We answered to all queries in the new version.

1. Author: Need university name here.
   Answer: The author affiliation is as: 2Department of Neurosurgery, Hackensack University Medical Center, New Jersey, USA.

2. Please include information on the patient’s ethnicity in both the abstract and the case presentation sections.
   Answer: Caucasian was added to abstract in the abstract and so in the site of j in abstract part.

3. Consent part: Has this been edited correctly?
   Answer: Yes it is correct.

4. Contributions:
   Answer: We corrected this part as following:
   FN, NB, and SY had major contribution in taking care of the patient, literature review and drafting of the manuscript. MEK has made substantial contribution to literature review, correction and final approval of the manuscript.

Thanks for your detailed corrections. Looking forward to hearing from you.

Regards

FN
MEK
NB
SY